Author’s response to reviews

Title: Depressive symptoms and early retirement intentions among Danish eldercare workers: Cross-sectional and longitudinal analyses

Authors:

Mette A. Nexø (man@nrcwe.dk)
Vilhelm Borg (vbo@nrcwe.dk)
Camilla S. Sejbæk (case@sund.ku.dk)
Isabella G. Carneiro (Isabella.Carneiro@phe.gov.uk)
Pernille U. Hjarsbech (Pehj@kora.dk)
Reiner Rugulies (rer@nrcwe.dk)

Version: 3 Date: 5 June 2015

Author’s response to reviews:

Dear Victorino Silvestre and Dr Yuan-Pang Wang

We deeply appreciate the opportunity to revise the manuscript “Depressive symptoms and early retirement intentions among Danish Eldercare workers: Cross-sectional and longitudinal analyses” (MS: 2063962700161475) by Nexo, et al. in BMC Public Health.

We have followed the reviewer’s constructive suggestions and in the enclosed files, we have made major revisions to the manuscript (the changes are highlighted in yellow). We resubmit this version of the manuscript along with a response to the reviewer’s comments.

We hope this manuscript has addressed the issues of concern and is suitable for publication in BMC Public Health.

We confirm that this manuscript has not been published elsewhere and is not under consideration to be published by another journal. All of the authors have approved the manuscript and agreed to its submission to BMC Public Health. Please do not hesitate to contact me if you have any queries regarding the manuscript.

Best regards,

Mette Andersen Nexø
(Corresponding author)
The National Research Centre for the Working Environment
Lersoe Parkallé 105, DK-2100 Copenhagen, Denmark
Email: man@nrcwe.dk
Telephone: +45 39 16 52 00